Author's response to reviews

Title: Limiting the Caesarean Section rate in low risk pregnancies is key to lowering the trend of increased abdominal deliveries: an observational study

Authors:

Ilse Delbaere (ilse.delbaere@ugent.be)
Hendrik Cammu (hendrik.cammu@uzbrussel.be)
Evelyne Martens (evelyne.martens@kindengezin.be)
Inge Tency (inge.tency@ugent.be)
Guy Martens (guy.martens@kindengezin.be)
Marleen Temmerman (marleen.temmerman@ugent.be)

Version: 5 Date: 8 December 2011

Author's response to reviews:

Dear Editor,

We edited table 3 into a table 3A and table 3B and changed citations within the text. Previously, there were two tables under table 5, which we merged into one table now. All vertical lines within tables are erased.

Yours sincerely,
Ilse Delbaere